Title: Long-term culture of cholangiocytes from liver fibro-granulomatous lesions

Authors:

Luciana B. Chiarini (chiarini@biof.ufrj.br)
Christina M. Takiya (takiyacm@hotmail.com)
Radovan Borojevic (radovan@histo.ufrj.br)
Alvaro N.A. Monteiro (monteian@moffitt.usf.edu)

Version: 4 Date: 30 March 2006

Keywords section - The keywords section was removed from the manuscript.

Section headings - The full stop was removed from the end of all headings and sub-headings in the manuscript.

Section headings - Sentence case was used for all headings and sub-headings in the manuscript.

Figure title - The figure titles were removed.

Figures - The figures were cropped to minimise white space around the image.